\documentclass{article}

\setlength{\parindent}{0pt}

\begin{document}

\thispagestyle{empty}

\centerline{\textbf{Math 165: Honors Calculus I \hfill Fall 1998}}
\centerline{\textbf{Course Information}}

\bigskip
INSTRUCTOR: Dennis M. Snow, 313 CCMB, 631-7597, e-mail: snow.1@nd.edu

\bigskip
CLASSROOMS: MWF 11:45 Pasquerilla 102; H 11:00 DBRT 131

\bigskip
OFFICE HOURS: 2--3 MWF, or by appointment

\bigskip
TEXT: T.~Apostol, Calculus, Vol.\ I, 2nd Ed.


\bigskip
EXAMS:  There will be two one-hour exams and a two-hour
final on the following dates:
\begin{center} \begin{tabular}{ll}
Exam I &Thursday, October 1\\
Exam II &Thursday, November 19\\
Final &Tuesday,  December 15, 8:00 A.M.\\
\end{tabular} \end{center}
Make-up exams will only be given for officially excused absenses.

\bigskip
QUIZZES: There will be a short quiz given on Thursday of every week,
except for exam weeks and the first/Thanksgiving week of classes. Each
quiz is worth 10 points.

\bigskip
ASSIGNMENTS: There will be regular assignments due on the class day after
they are assigned (Wednesday's assignment is due on Friday).  Points will
be deducted for late assignments and may not be accepted at all.

\bigskip
ATTENDANCE: You are expected to attend every class. Excessive
absenses may result in lowering your grade and even failing the
course.

\bigskip
HONOR CODE: The Honor Code is in effect for all exams, quizzes and
assignments. You are encouraged to work together on the assignments,
but copying in any form or submitting work done by others as your own
is a violation of the Honor Code.  A group of up to three students may
submit an assignment together, as long as the group members are clearly
identified and they all contribute to all parts of the assignment.

\bigskip
EMAIL: Information about the course and regular grade updates will be
sent to you by e-mail. If you have not already done so, you should get
your user-id and password from the Information Resource Center, first
floor of the Comp/Math Bldg. The IRC will also help you get started with
e-mail. I encourage you to ask questions through this medium; you will
always recieve a written reply.

\bigskip
GRADES:  Grades will be based on a total of 550 points broken down as follows:
\begin{center} \begin{tabular}{lcl}
Exam I\&II &2 @ 100  &= 200\\
Final &&= 150\\
Quizzes &10 @ 10 &= 100\\
Assignments &(scaled) &= 100\\
\end{tabular} \end{center}

\end{document}
